Author’s response to reviews

Title: Long-term renal graft outcome after parathyroidectomy- a retrospective single centre study

Authors:

Margret Patecki (patecki.margret@mh-hannover.de)

Irina Scheffner (Scheffner.Irina@mh-hannover.de)

Hermann Haller (Haller.Hermann@mh-hannover.de)

Wilfried Gwinner (Gwinner.Wilfried@mh-hannover.de)

Version: 2 Date: 05 Feb 2020

Author’s response to reviews:

Dr. Margret Patecki Hannover, 5th of January 2020

Medical School Hannover

Carl-Neuberg-Straße 1

30625 Hannover, Germany

BMC Nephrology

BioMed Central

Floor 6, 236 Gray's Inn Road

London

WC1X 8HB

United Kingdom
Dear Editors and Reviewers,

Thank you so much for considering our manuscript for publication in BMC Nephrology.

Here are the answers to your comments:

Editor Comments:

1. Under your Ethics approval and consent to participate heading in the Declarations section, please clarify whether informed consent obtained from participants was written or verbal. If verbal, please state the reason and whether the ethics committee approved this procedure.

Answer: The consent obtained from participants was written. We supplemented this at page 14, line 325.

2. Thank you for providing an Availability of data and materials statement for the Declarations section. Please note that this section refers to where the raw data generated/analysed during your study can be found. This does not refer to data that has been processed and summarized in tables/figures. Therefore, if your paper does include all the raw data, please do not change your statement. If it does not, please further clarify where the raw data can be found.

   If you wish to share the data through correspondence then please include the statement “The datasets used and/or analysed during the current study are available from the corresponding author on reasonable request.”

   If you do not wish to share your data, please state this along with the reason.

   Answer: We want to share the data through correspondence and included this statement at page 14, lines 331-333.

3. When submitting your revised manuscript please ensure you do so as a single clean copy without any tracked changes, colored or highlighted text, as these are no longer required at this stage of the editorial process.
Answer: We now submit the revised manuscript as clean copy.

We are looking forward to hearing from you.

Yours sincerely,

M. Patecki and W. Gwinner